Author's response to reviews

Title: Pre-operative intraocular pressure does not influence outcome of trabeculectomy surgery: a retrospective cohort study

Authors:

Nisha Nesaratnam (nn252@cam.ac.uk)
Nicholas Sarkies (nicholas.sarkies@addenbrookes.nhs.uk)
Keith R Martin (krgm2@cam.ac.uk)
Humma Shahid (hummashahid@hotmail.com)

Version: 4 Date: 24 September 2014

Author's response to reviews:

(1) Request for ethics statement - this study was a retrospective notes review, and did not involve human subjects. Research was done in accordance with the Declaration of Helsinki, and this has been written in the Methods section.

(2) Request for consent statement - as this was a retrospective notes review, no consent for participation was required.

(3) Request for Line Numbering - line and page numbers have been included.